Author's response to reviews

Title: Impact of sample acquisition and linear amplification on gene expression profiling of lung adenocarcinoma: laser capture micro-dissection cell-sampling versus bulk tissue-sampling

Authors:

Eric W Klee (klee.eric@mayo.edu)
Sibel Erdogan (erdogan.sibel@mayo.edu)
Lori Tillmans (tillmans.lori@mayo.edu)
Farhad Kosari (kosari.farhad@mayo.edu)
Zhifu Sun (sun.zhifu@mayo.edu)
Dennis Wigle (wigle.dennis@mayo.edu)
Ping Yang (yang.ping@mayo.edu)
Marie Christine Aubry (aubry.mariechristine@mayo.edu)
George Vasmatzis (vasmatzis.george@mayo.edu)

Version: 4 Date: 29 January 2009

Author's response to reviews: see over
BMC Medical Genomics
Editorial Board,

Thank you for taking the time to review this manuscript. We are pleased you have accepted for publication. The requested modification by Reviewer #2 has been made. Additionally, we have revised the Abstract per your editorial board’s request.

Please let us know what the next steps are towards getting the article published.

Sincerely,

Eric William Klee
Stable 3-15
Mayo Clinic
Rochester, MN 55905
507-266-5741 (phone)
507-266-5193 (fax)
Klee.eric@mayo.edu